\documentclass[twoside]{comflowpore}
\begin{document}

\begin{frontmatter}

\title{This is the title}   

\author[label1]{F. Lastname}
\author[label1,label2]{F. Lastname}
\author[label2]{F. Lastname}

\address[label1]{adress 1}
\address[label2]{adress 2}

\begin{keyword}  %%% KEYWORDS..
	keyword1 \sep keyword2 \sep keyword3  
\end{keyword}

\end{frontmatter}

%%%%%%%%%%%%%% The abstract begins here %%%%%%%%%%%%%%%%%%%%%%ù

\section{Introduction}

Example of bibliographic citations:
\cite{article1},\cite{article5},\cite{book6} or \cite{article1,article5}

\section{A new section}

An equation:
\begin{equation}
\label{eq1}    % Give a unique label
a^2+b^2=c^2
\end{equation}
One cite an equation like this:  eq. \eqref{eq1}.

Table example:
\begin{table}[h]
\begin{center}
\begin{tabular}{lll}
\hline\noalign{\smallskip}
first & second & third  \\
\noalign{\smallskip}\hline\noalign{\smallskip}
number & number & number \\
number & number & number \\
\noalign{\smallskip}\hline
\end{tabular}
\caption{Please write your table caption here}
\label{tab:1}       % Give a unique label
\end{center}
\end{table}

Figure example:

\begin{figure}[h]
\begin{center}
   \includegraphics[width=0.5\hsize]{figure5.eps}   % file with extension  eps / jpg / tiff / png / pdf
   \caption{Please write your figure caption here}
   \label{fig:1}      % Give a unique label
\end{center}
\end{figure}

A reference to  Fig. \ref{fig:1}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% Bibliography %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{thebibliography}{}

\bibitem{article1}
% Format for Journal Reference
 Author4, Article title, Journal, Volume, page numbers (year)
\bibitem{article5}
% Format for Journal Reference
 Author5, Article title, Journal, Volume, page numbers (year)
% Format for books
\bibitem{book6}
 Author6, Book title, page numbers. Publisher, place (year)

\end{thebibliography}

\end{document}

